Author's response to reviews

Title: Do expanded seven-day NHS services improve clinical outcomes? Analysis of comparative institutional performance from the "NHS Services, Seven Days a Week" project 2013-2016

Authors:

Hoong-Wei Gan (hoong.gan.11@ucl.ac.uk)
Danny Wong (danny.jn.wong@gmail.com)
Benjamin Dean (bendean1979@gmail.com)
Alistair Hall (a.s.hall@leeds.ac.uk)

Version: 1 Date: 04 Dec 2016

Author's response to reviews:

Editor-In-Chief
BMC Health Services Research
28 November 2016

Dear Editor,

Re: Re-submission of Research Article: Do expanded seven-day NHS services improve clinical outcomes? Analysis of comparative institutional performance from the “NHS Services, Seven Days a Week” project 2013-2016

Further to your letter dated 8 November 2016 inviting revision and resubmission of our manuscript, we have pleasure in enclosing our revised draft for your consideration for publication in BMC Health Services Research and a detailed point-by-point response to the reviewer’s comments.

As requested, we have resubmitted a clean version of the revised manuscript. The authors' detailed response letter has been included as a supplementary file.

We continue to trust you will find this article of timely significance and look forward to your reply.
Many thanks.

Yours sincerely,

Dr. Hoong-Wei Gan BMedSci (Hons) BM BS MSc MRCPCH
Clinical Research Fellow/ Paediatric Endocrinology Specialty Registrar
UCL Institute of Child Health & Great Ormond Street Hospital for Children NHS Foundation Trust

Dr. Danny Wong BSc MBBS AKC MRCP FRCA
Research Fellow/ Anaesthesics Specialty Registrar
UCL National Institute of Academic Anaesthesia Health Services Research

Mr. Benjamin Dean BM BCh BA DPhil MRCS(Ed)
Research Fellow/ Orthopaedic Registrar
Nuffield Department of Orthopaedics, Rheumatology & Musculoskeletal Sciences

Prof. Alistair Hall MBChB PhD FRCP
Professor of Cardiovascular Epidemiology/ Consultant Cardiologist
Leeds MRC Medical Bioinformatics Centre